Author’s response to reviews

Title: Nursing students' transfer of learning outcomes from simulation-based training to clinical practice: A focus-group study

Authors:

Jørn Hustad (jorn.hustad@uia.no)
Berit Johannessen (berit.johannessen@uia.no)
Mariann Fossum (mariann.fossum@uia.no)
Olav Johannes Hovland (johannes.hovland@uia.no)

Version: 2 Date: 09 Oct 2019

Author’s response to reviews:

Editor's comments are addressed and question number 1-6 has been addressed.

The interview guide questions are included in the text of the manuscript. COREQ is removed and all tracked changes are removed from the manuscript.

Thank you,

Jørn Hustad